**Author's response to reviews**

**Title:** Undergraduate medical education in emergency medical care: A nationwide survey at German medical schools

**Authors:**

Stefan K Beckers (sbeckers@ukaachen.de)
Arnd Timmermann (atimmer@zari.de)
Michael Müller (mp-mueller@web.de)
Matthias Angstwurm (matthias.angstwurm@med.uni-muenchen.de)
Felix Walcher (walcher@trauma.uni-frankfurt.de)

**Version:** 7  **Date:** 6 May 2009

**Author's response to reviews:**

Cover Letter to Manuscript ID 4421879192326577
Undergraduate medical education in emergency medical care: A nationwide survey at German medical schools
Stefan K. Beckers; Arnd Timmermann; Michael P. Müller; Matthias Angstwurm and Felix Walcher

Dear Editor,

We made all the required major and minor changes mentioned in the mail from May 6th 2009, and finally proofed the revised manuscript at this stage.

Thank you very much again for your kind consideration in this matter.

Best regards

Stefan K. Beckers

Section Emergency Medical Care, Department of Anaesthesiology and AIXTRA – Aix-la-Chapelle Centre for Interdisciplinary Training in Medical Education
Medical Faculty RWTH Aachen University, Pauwelsstr. 30, D – 52074 Aachen, Germany
e-mail: sbeckers@ukaachen.de
Phone: +49 (0) 241 – 80 89974
FAX: +49 (0) 241 – 80 82304